Author's response to reviews

Title: Metropolitan-level ethnic residential segregation, racial identity, and body mass index among U.S. Hispanics: a multilevel cross-sectional study

Authors:

   Kiarri N Kershaw (k-kershaw@northwestern.edu)
   Sandra S Albrecht (ssalb@live.unc.edu)

Version: 2 Date: 3 April 2013

Author's response to reviews:

Dear Dr. Pafitis,

Please find an updated version of our manuscript, "Metropolitan-level ethnic residential segregation, racial identity, and body mass index among U.S. Hispanics: a multilevel cross-sectional study." A sentence stating that the BRFSS survey was approved by the Institutional Review Boards of the Centers for Disease Control and Prevention and all participating states' Departments of Health has now been included in the Methods section.

Thanks and we look forward to your decision,

Kiarri